Author’s response to reviews

Title: Diffuse Cavernous Hemangioma of the skull misdiagnosed as skull metastasis in breast cancer patient: one case report and literature review

Authors:

Huizhi Liu (lihuizhifl@163.com)

Xiaojing Chang (xiaojinge-mail@163.com)

Hua Shang (fangliaoke1@163.com)

Feng Li (lifengdoc@163.com)

Huandi Zhou (515499163@qq.com)

xiaoying Xue (xxy0636@163.com)

Version: 4 Date: 21 Jan 2019

Author’s response to reviews:

Our responses:

-- If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

We thank the editor’s suggestion, and have added abbreviations in the Declarations.

--Please remove the patient's details in the supplementary images (DOB,patient ID etc)

The patient's details (DOB,patient ID etc) have been removed.

-- Please ensure that the email address for the corresponding author matches that which has been listed on our editorial system- currently they differ.

We are so sorry for this mistake, there are two emails for the corresponding author, we have added another in e-mail addresses of authors of page 1.

-- Please also add a section “Supplementary files” where you list the following information about your supplementary material:
We fully understand the editor’s concern, and have added these information in “Supplementary files” section.

Please ensure that all additional files are explicitly referred to in the main text.

All additional files were referred to in the main text.

-- Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

We fully understand the editor’s concern, and have revised the author's contributions.

CX participated in the treatment of the patient, analyzed of aggregated data and drafted this manuscript. ZH participated in the acquisition of data and revision of the manuscript. XXY participated in the critical review and substantively revised it.

All authors listed in this manuscript have made substantive contributions.

-- Thank you for providing a response to the reviewers/CARE checklist. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.

The response to the reviewers/CARE checklist have been removed. Thank you!

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We thank the editor, and have uploaded our manuscript as a single, final, clean version. Thank you!